



Article EP-ADTA: Edge Prediction-Based Adaptive Data Transfer Algorithm for Underwater Wireless Sensor Networks (UWSNs)

Bin Wang *¹⁰, Kerong Ben, Haitao Lin, Mingjiu Zuo and Fengchen Zhang

College of Electronic Engineering, Naval University of Engineering, Wuhan 430033, China; benkerong08@163.com (K.B.); figue2015@163.com (H.L.); zuomingjiu@126.com (M.Z.); fczhang0606@163.com (F.Z.)

* Correspondence: wbnavy@163.com

Abstract: The underwater wireless sensor network is an important component of the underwater three-dimensional monitoring system. Due to the high bit error rate, high delay, low bandwidth, limited energy, and high dynamic of underwater networks, it is very difficult to realize efficient and reliable data transmission. Therefore, this paper posits that it is not enough to design the routing algorithm only from the perspective of the transmission environment; the comprehensive design of the data transmission algorithm should also be combined with the application. An edge prediction-based adaptive data transmission algorithm (EP-ADTA) is proposed that can dynamically adapt to the needs of underwater monitoring applications and the changes in the transmission environment. EP-ADTA uses the end–edge–cloud architecture to define the underwater wireless sensor networks. The algorithm uses communication nodes as the agents, realizes the monitoring data prediction and compression according to the edge prediction, dynamically selects the transmission route, and controls the data transmission accuracy based on reinforcement learning. The simulation results show that EP-ADTA can meet the accuracy requirements of underwater monitoring applications, dynamically adapt to the changes in the transmission environment, and ensure efficient and reliable data transmission in underwater wireless sensor networks.

Keywords: underwater wireless sensor networks (UWSNs); underwater monitoring applications; edge computing; intelligent routing algorithm; reinforcement learning (RL); auto regressive and moving average (ARMA)

1. Introduction

Underwater wireless sensor networks are widely used in early disaster warning, pollutant monitoring, hydrological data monitoring and collection, marine resource exploration, auxiliary navigation, and marine military applications [1], and have become an important infrastructure component of the underwater Internet of Things [2]. Underwater wireless sensor networks consist of two parts, a surface wireless sensor network and an underwater wireless sensor network, and include three-dimensional (3D) mobile, fixed, or mixed networks. Data are collected by underwater sensor nodes, which can be transmitted from underwater to surface sink nodes in the form of hop-by-hop forwarding through communication nodes at different depths, or collected by AUV and then sent to the surface. The surface network part uses a radio to realize the information transmission between the underwater network and the shore-based data center. Although underwater data can be transmitted by radio, wireless light, sound, and other media, underwater long-distance data transmission still relies on the underwater acoustic mode.

Compared with terrestrial wireless sensor networks based on RF communication, underwater wireless sensor networks are characterized by an unstable, uneven transmission environment; many interference factors; a small transmission channel capacity; a large delay; low reliability; limited energy of communication nodes; a limited energy supply; and a high



Citation: Wang, B.; Ben, K.; Lin, H.; Zuo, M.; Zhang, F. EP-ADTA: Edge Prediction-Based Adaptive Data Transfer Algorithm for Underwater Wireless Sensor Networks (UWSNs). *Sensors* 2022, 22, 5490. https:// doi.org/10.3390/s22155490

Academic Editors: Jaime Lloret and Abdallah Makhoul

Received: 11 June 2022 Accepted: 20 July 2022 Published: 23 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). degree of space-time dynamics [3,4]; as a result of these characteristics, it is very difficult to realize underwater wireless network communication. At present, the underwater wireless channel is considered to be one of the most difficult-to-manage channels. In order to overcome the adverse effects of the underwater transmission environment, researchers have divided the protocol stack of underwater wireless sensor networks into a physical layer, data link layer, network layer, transmission layer, and application layer, and carried out cross-layer optimization design. They have adopted the methods of channel awareness [5,6], energy consumption awareness [7–10], and delay awareness [5,11] to better select transmission channels. Layered [9,12] or clustering [12–15] methods have been adopted to divide the underwater network, balance the energy consumption, and extend the network life. Previous studies have endowed the transmission routing with intelligence, so that it can adapt to the changes in the transmission environment; optimize the transmission quality; bypass network holes [16,17]; and realize efficient, reliable, and stable data transmission. However, due to the complex, dynamic, and extremely limited underwater transmission environment, when devising underwater monitoring applications having a stable and huge data transmission demand, it is not enough to only pay attention to the design and combination optimization of the physical layer, link layer, and network layer of the routing protocol. The uncontrolled demand for transmission of a large quantity of high-precision monitoring data will make the routing protocols ineffective due to congestion and energy depletion. The cross-layer design of routing protocols, which is not closely combined with the application layer, cannot maximize the use of transmission resources and achieve effective data transmission [18,19]. It is necessary to design an adaptive data transmission method that can dynamically adjust the accuracy of data transmission according to the transmission conditions and application requirements while correctly selecting the relay node.

End–edge–cloud architecture [20,21], which is a hierarchical architecture based on terminal equipment, the edge network, and the cloud environment, has been gradually applied to the Internet of Things system. By replacing centralized cloud computing with edge computing dispersed in the edge network, this approach can better meet the needs of delay-sensitive applications, reduce data transmission load, and improve the overall performance of the network [22]. The underwater wireless sensor network is also applicable to the end-edge-cloud architecture. As a result of the improvement in the computing power of underwater nodes, edge computing will play a greater role in improving the performance of underwater wireless sensor networks [23,24]. Edge computing has been applied to underwater mobile wireless sensor networks composed of AUV and sensor nodes to reduce the data transmission burden between sensor nodes and the AUV, and between the AUV and shore-based data centers [24,25]. In the fixed network, the sensor nodes used for data collection can be defined as the information end; the wireless communication network is composed of underwater sensor nodes; communication nodes and sink nodes form the edge network; the shore-based data center network comprises the cloud environment; and the sink node is used as the gateway node connecting the underwater edge network and the cloud environment. During the process of underwater monitoring, the data are collected by the underwater sensor nodes, forwarded to the surface sink nodes via each hop through the underwater communication nodes, and then transmitted to the shore-based data center via satellite. The continuous data transmission demand for underwater monitoring applications has led to a more serious burden on the previously limited underwater wireless communication. However, as time series data and monitoring data generally show a great correlation and redundancy in time and space [26,27], the new monitoring data can be predicted in the edge network using an edge prediction algorithm and historical data. After edge prediction, only the prediction parameters and correction data that meet the application accuracy requirements should be transmitted, which can effectively compress the actual data transmission volume, thereby reducing the transmission delay and energy consumption, improving the packet-sending success rate, and enhancing the communication performance of underwater wireless sensor networks.

Agent modeling is a method used to divide and conquer complex systems. Intelligent agents can respond and process in a collaborative manner based on the intelligent policies by understanding the environment and other agent statuses [28,29]. The communication of the underwater wireless sensor network is complex and difficult. Each node in the edge network can be defined as an intelligent agent [30]. Through cooperation and communication, intelligent agents can effectively improve the performance of underwater wireless network communication [31,32]. Mobile agents have been used to enhance the network scalability [12]. Multiple agents have been used to optimize routing selection [28,29] and to improve the rationality of resource allocation [30,31]. Reinforcement learning is an important method for improving agent intelligence. It allows agents to learn the optimal strategy through trial-and-error training. As a result of the changes in the transmission channel, transmission node performance, and application transmission requirements, the agent must dynamically adjust the routing strategy and accuracy of data transmission according to the environmental state to ensure the performance of underwater communication. Over a certain period, reinforcement learning can equip agents with this ability [28,29]. Therefore, the data transmission strategy is based on reinforcement learning, which incorporates the combination of the environmental status and business requirements into the state space, and selects the best data transmission accuracy while selecting the best relay node, to obtain a data transmission strategy. Under the requirements of underwater monitoring data accuracy, the number of transmitted correction data will be dynamically adjusted according to the change in the transmission environment. When the transmission environment is good, more correction data will be transmitted to ensure higher transmission accuracy. When the transmission environment is poor, the number of transmitted correction data can also be dynamically reduced to ensure a higher packet-transmission success rate. When selecting the best relay node, the data transmission strategy comprehensively considers the location of neighbor nodes, residual energy, and transmission channel delay. In order to avoid too many nodes participating in reinforcement learning and improve the convergence speed of reinforcement learning, a virtual pipeline is built between the source node and the target node based on the residual energy, and the relay node is only selected from the virtual pipeline. The environment and neighbor agent statuses are updated by the monitor and control channel established based on the broadcast between neighbor nodes. The agent based on reinforcement learning can independently find the best data transmission strategy according to the change in the environmental state, and realize efficient and reliable underwater wireless network communication.

In the end–edge–cloud and agent modeling architecture, with the comprehensive design of all function layers of underwater network, this paper presents an adaptive data transmission algorithm (EP-ADTA) based on edge prediction and reinforcement learning. The main contributions include the following:

(1) The end–edge–cloud architecture is used to define the fixed three-dimensional underwater wireless sensor network, and edge computing is applied to the data transmission of the underwater wireless sensor network. A new method of realizing the efficient data transmission of the underwater wireless sensor network by replacing communication with computing is proposed.

(2) Taking underwater time series data monitoring as an example, a set of edge predictions based on the ARMA model is proposed to predict the new monitoring data at the edge network. Edge prediction cannot only automatically match the optimal prediction parameters, but also dynamically adjust the amount of correction data to be transmitted according to the accuracy changes in the application data.

(3) Based on agent modeling technology, a data transmission algorithm based on reinforcement learning is proposed. The intelligent agent located in the edge network can sense the changes in the transmission channels' delay, the neighbor nodes' location and residual energy, automatically select the best transmission route, and adapt the data transmission accuracy to ensure efficient and reliable data transmission in underwater wireless sensor networks.

(4) The simulation results show that the edge prediction algorithm can meet the accuracy requirements of underwater time series data monitoring, and the data transmission algorithm based on reinforcement learning can dynamically adapt to the changes in the transmission environment, which can greatly reduce the data transmission delay and energy consumption of underwater wireless sensor networks, and effectively improve the packet-sending success rate.

The remainder of this paper is organized as follows: Section 2 introduces the related research work; Section 3 introduces the framework of the edge prediction-based adaptive data transmission algorithm (EP-ADTA); Section 4 introduces the edge prediction algorithm based on ARMA (EP-ARMA); Section 5 introduces the data transmission algorithm based on reinforcement learning (RL-ADTA); Section 6 introduces the simulation experiment and performance analysis; Section 7 is the conclusion.

2. Related Works

The complexity and time variation of the underwater network environment have encouraged researchers to study efficient and reliable data transmission methods from different perspectives. The routing protocol is the key to realizing data transmission in underwater wireless sensor networks. According to the implementation method, the routing protocol is divided into the geographic routing protocol [33,34] and the opportunistic routing protocol [7,16]. The geographic routing protocol uses the relative or absolute location information of underwater nodes to design the packet-forwarding path, avoiding a large amount of broadcast information caused by the routing search [35]. The opportunistic routing protocol solves the problem of intermittent connection caused by mobility and the dynamic underwater acoustic link in sparse acoustic wireless networks, and uses 'storecarry-forward' opportunistic routing to maximize packet forwarding. According to the guaranteed type of quality of services (QoS), routing protocols are divided into those that focus on energy consumption balance, transmission delay, network life, or other indicators. Routing protocols can also be divided into traditional routing protocols and intelligent routing protocols according to whether artificial intelligence algorithms are used [3,36]. Traditional routing protocols do not need data learning and model optimization. In specific network scenarios, the optimization effect of specific indicators is good, but the network adaptability is poor, and there is only one optimization goal. The intelligent routing protocol can collect the network status data, form awareness of the network environment through data learning, support the dynamic adjustment of routing, and realize the balanced optimization of multiple QoS indicators, and has strong network adaptability [36,37]. Researchers have also adopted a cross-layer design method to further improve the performance of the routing protocol through the joint design of data link layer, transport layer, and application layer protocols [10,38]. Several typical data transmission algorithms and their characteristics are listed in Table 1. The communication efficiency mainly reflects the ability of the routing protocol to send effective information and control the transmission delay and energy consumption.

RLOR uses opportunistic routing to improve the reliability of routing. The basis of RLOR's selection of relay nodes is the use of transmission direction and distance, channel quality, number of neighbors, and residual energy of relay nodes to ensure that the best route can be dynamically selected. RCAR takes more account of the congestion avoidance method of relay nodes. In RCAR, the transmission delay and energy consumption distribution in underwater data communication are optimized. MCR-UWSN divides the network into multiple clusters; dynamically selects cluster heads based on the optimized heuristic algorithm; and constructs inter-cluster routes to achieve efficient data transmission. KACO adopts clustering and heuristic algorithms for network layering and clustering. PB-ACR can distinguish the priority of different applications. In PB-ACR, high- priority applications choose the shortest route, and low-priority applications choose the route with more residual energy. ACOR supports the adaptive adjustment of data sampling frequency according to data correlation to reduce transmission energy consumption. Although both RLOR and

RCAR can provide good QoS under extremely limited transmission conditions, it is not enough to simply consider overcoming the problems of the underwater environment and seek the best transmission route. Dynamic network traffic will seriously affect the performance of routing protocols. These protocols also need to be combined with applications to achieve efficient information transmission, rather than just data transmission. The use of layers, clusters, and inter-cluster routing based on the heuristic algorithm has improved the balance of energy consumption of MCR-UWSN and KACO algorithms; however, there is no detailed description of the performance change in algorithms under different network traffic conditions. Although PB-ACR provides differentiated routing for different priority services, it does not reduce the overall energy consumption of the network. Although ACOR can improve communication efficiency by sampling service data, the sampling technology is not suitable for applications having a large number of irregular and abrupt data transmission requirements, and ACOR does not address the problem of abnormal data.

Table 1. Several typical routing protocols for underwater wireless sensor networks.

Algorithm	Network	Route Establishment Method	Application Content	Communication Efficiency
VBF [33], CARP [6]	3D Network	Location vector and neighbor information	No distinction	Low
RLOR [16], RCAR [5]	3D Network	Delay, energy consumption and reliability; establish routing based on reinforcement learning algorithm	Traffic	Middle
MCR-UWSN [14], KACO [12]	3D Network	Energy, depth; layer or cluster; establish routing based on heuristic algorithm	No distinction	Middle
PB-ACR [39], ACOR [40]	3D Network	Energy consumption; establish routing based on ant colony algorithm	Prioritization, data relevance	Middle
SDA [41], DBP [42]	3D Network	neighbor information	Content fusion and prediction	Middle
TBDP [25]	AUV auxiliary network	Carry transmission data based on AUV	Content prediction	Middle
EP-ADTA	3D Network	Delay and energy consumption, and establish routing based on reinforcement learning	Content prediction and fusion	High

SDA is used for event detection and source determination. When the sensor's sensing data are transmitted upward along the routing tree, SDA comprehensively determines whether the underwater network generates abnormal data by comparing the sensing data of multiple adjacent sensor nodes with the normal sensing range. The neighboring sensor nodes only collect and forward their sensing data to the receiving node when these sensor nodes jointly detect possible events. This strategy avoids the unnecessary routing of perceptual data that is unfavorable to event detection, reduces energy consumption, and increases network capacity. However, this fusion can save energy for the network only when it is used to judge abnormal conditions. DBP is an event detection method based on linear prediction. The core idea of DBP is to use a simple model to capture the trend in data changes, and use flexible rules to calculate and deal with the existing interference. One of the limitations of DBP is its low prediction accuracy, which is suitable for short-term linear trend data. Although both SDA and DBP are based on realizing the communication of sensing data and can also be applied to underwater 3D networks, they focus on the detection of events, and do not address the construction of transmission routes and the methods for delivering data from sensor nodes to sink nodes in detail.

TBDP is a two-level bidirectional prediction model that is suitable for 3D underwater wireless sensor networks with an AUV, and realizes communication by relying on the data carried by the AUV. TBDP defines the AUV auxiliary network as the end–edge–cloud system, in which the underwater sensor system including the sensor node and the communication gateway node is defined as the end system, the AUV and the water surface sink node are defined as the edge system, and the shore-based data center is defined as the cloud environment [43,44]. Two-way prediction algorithms are set between the end system and the edge system, and between the edge system and the cloud environment, to compress the transmission data demand. Although using an AUV as a method of underwater data acquisition has certain advantages, the real-time performance of data acquisition is still insufficient compared with that of the fixed 3D underwater sensor network. In addition, the prediction algorithm proposed by TBDP has room for improvement in prediction accuracy, applicability, and mobility for different application requirements.

Therefore, on the basis of the above research, in order to improve the communication efficiency of the fixed 3D underwater wireless sensor network, it is necessary to carry out research on the adaptive data transmission algorithm that comprehensively considers business applications and the transmission environment. The algorithm realizes data compression or preprocessing based on edge prediction, and realizes the selection of relay nodes and the control of data transmission accuracy based on reinforcement learning.

Underwater environment monitoring includes the monitoring of the underwater temperature, salinity, pressure, PH value, dissolved oxygen, NH4-N, and other time series data, and various applications have great differences in terms of the quantity, accuracy, frequency, speed, and reliability of data transmission. However, in this paper, only the underwater temperature monitoring is taken as an example to illustrate the system framework and algorithm. Other types of monitoring applications can be improved in the system framework in combination with the application characteristics.

3. System Architecture

3.1. End–Edge–Cloud Architecture

The fixed 3D underwater wireless sensor network is composed of shore-based/ship data centers, buoy sink nodes, underwater floating sensor/communication nodes, and bottomed sink sensor nodes [3,45]. Both the bottom and floating sensor nodes have data acquisition capabilities. Based on the wireless communication mode, the data are sent to the underwater communication nodes, and then forwarded via each hop through the underwater communication nodes to send the monitoring data to the surface sink nodes, and then sent by the sink nodes to the data center via satellite. End-edge-cloud architectures include the cloud environment, edge network, and information end. In the Internet of Things, the information terminal is distributed in the monitoring environment and is responsible for data collection; the edge network provides data access channels for various information terminals and is responsible for data transmission; and the cloud environment gathers all kinds of data, is responsible for centralized data processing, and provides services for upper layer applications. Therefore, the sensor nodes responsible for data collection are defined as the information end; the underwater wireless communication network composed of the underwater sensor, communication, and sink nodes is defined as the edge network; and the broadband wireless communication network composed of shore- and ship-based data centers is uniformly defined as the cloud environment; this is shown in Figure 1.

In the underwater monitoring application, the data obtained by the information terminal are transmitted via each hop through the edge network and continuously sent to the cloud environment. The change trend, acquisition accuracy, and outliers of monitoring data are very important for monitoring applications, but the huge quantity of transmitted data will pose large problems for the edge network. As the delay increases, the energy consumption increases, and the packet-sending success rate decreases. Therefore, before the data enters the edge network, the network throughput should be reduced and the

data transmission performance improved by preprocessing to realize the transmission data compression. Although data preprocessing must allocate the computational energy consumption for underwater network nodes, the computational energy consumption is often far less than the communication energy consumption. According to [46], the energy consumed by executing 3000 instructions is equivalent to transferring 1 bit of data a distance of 100 m. With the rapid improvement of computing chip technology and computing modes [47], the gap between computing energy consumption and communication energy consumption will continue to widen.



Figure 1. End-edge-cloud architecture of underwater wireless sensor networks.

The data collected by underwater three-dimensional monitoring belongs to time series data, which have certain continuity in space and time, and it is possible to reduce the amount of data transmission through preprocessing. There are two methods to realize front-end monitoring data compression, i.e., using the adaptive sampling method or the time series data prediction algorithm. The frequency of adaptive sampling is closely related to the frequency of sampling data. If the sampling frequency is small, key data will be missed. If the sampling frequency is large, the effect of data compression will be lost. When the monitoring data changes dramatically in a short time, the adaptive sampling may not be able to capture the key data. Time series data prediction is based on historical data and data modeling to realize the learning and fitting of monitoring data, which can better reflect the change trend of the monitoring data. The parameters of the prediction model can be used to replace the actual monitoring data, and the data compression can be better realized. At the same time, selecting correction data of different threshold ranges can effectively adjust the accuracy of the monitoring application and avoid key data having drastic changes being missed. The surface sink node recovers the monitoring data according to the received prediction parameters and correction data based on the edge prediction algorithm.

3.2. System Composition

The nodes in the fixed 3D underwater wireless sensor network have certain computing and storage capabilities. Based on the underwater acoustic mode, they receive and send service data, and learn the status of the underwater transmission environment by monitoring the communication of the neighbor nodes. The monitoring data are collected from the sensor nodes, and forwarded by the communication nodes to the sink nodes. Therefore, each node constituting the edge network can be defined according to the agent, as shown in Figure 2.



Figure 2. Agent model of underwater wireless sensor network.

The agent in the edge network includes the communication module, edge prediction module, and intelligent data transmission module, and has an application requirements base, environment status base, transmission policy base, and service data base. The edge prediction module is only required for underwater sensor nodes and sink nodes. The application requirements base stores the requirements for data collection type, frequency, and accuracy of monitoring applications. The environment status base stores the state information of neighbor nodes, transmission channels, and the current node. The transmission policy base stores super-parameters, thresholds, and the V values [48]. The service data base stores the monitoring data, newly received prediction parameters, correction data, etc., and temporarily stores the sent data. The communication module is responsible for data transmission and receiving the packets. The edge prediction module can not only automatically match the optimal prediction parameters, but also generate monitoring prediction data based on the prediction model. The intelligent data transmission module is responsible for selecting the appropriate transmission route and data transmission accuracy according to the status of the environment. When the status update of the neighbor node is detected, the communication module is responsible for updating the content of the environment status base. When new data are received, the communication module receives the data; obtains the prediction parameters, correction values, and data source node information; and writes the new data into the service data base. When sending data, the intelligent data transmission module generates the best data transmission strategy and sends the data according to the selected route and data accuracy. When the communication module finds that the neighbor node has sent data, the intelligent data transmission module deletes the temporary data in the service database.

3.3. Operation Flow

In order to achieve efficient and reliable data transmission in underwater wireless sensor networks, cooperation between agents located in the edge networks is required. According to the location and function of the agent in the edge network, the agents are divided into the sensor agent, communication agent, and sink agent, corresponding to the underwater sensor node, communication node, and sink node, respectively.

The sensor agent should complete the following actions: (1) carry out data acquisition and edge prediction, and generate prediction parameters and correction data; (2) assemble data packets, generate adaptive transmission policies, and send the packets to relay nodes according to the policies; and (3) receive the packets and update the status of transmission environment.

The communication agent should complete the following actions: (1) repackage the data, generate an adaptive transmission strategy, and send the packet to the relay node according to the strategy; (2) receive the packets, update the status of transmission environment, and obtain the data.

The sink agent should complete the following actions: (1) receive data packets and obtain transmission data; (2) generate the new monitoring data based on edge prediction, historical data, received prediction parameters, and correction data, and then send them to the cloud environment.

The functions of the sensor agent, communication agent, and sink agent are composed of four basic processes, namely: (1) the data acquisition and edge prediction process; (2) the data transmission process; (3) the data receiving process; (4) and the edge prediction and data recovery process. Of these, the sensor agent needs to implement processes (1), (2) and (3); the communication agent needs to implement processes (2) and (3); and the sink agent needs to implement processes (3) and (4).

Process (1) is shown in Figure 3a. Based on the ARMA [49] model, it can automatically match the optimal prediction parameters, generate prediction parameters and correction data according to historical data, and provide the threshold of the maximum allowable error for the monitoring application.

Process (2) is shown in Figure 3b. The reinforcement learning calculates the new Q value [48] according to the status of neighbor nodes, transmission channels, and their own nodes, in addition to the V value, threshold and super-parameters, and finds the best transmission strategy, that is, the next hop node and the number of correction data. Then, the packets are generated according to the packet format and broadcast. When the sender hears that the data packet has been forwarded by the neighbor node, the sent data are deleted; otherwise, the Q value is calculated and retransmitted.

Process (3) is shown in Figure 3c. By monitoring and receiving the new data in the transmission environment, the status of the neighbor nodes and the transmission channels are updated. If this node is a relay node, the new data are received.

Process (4) is shown in Figure 3d. It is only used in the sink node and is implemented based on the ARMA model. It is used to restore the generated monitoring data according to the received prediction parameters, historical data, and correction parameters, and to send the generated monitoring data to the cloud environment.

Next, we introduce the edge prediction algorithm based on EP-ARMA and the adaptive data transmission algorithm based on RL-ADTA.



Figure 3. Operation flow of the agents: (**a**) data acquisition and edge prediction process; (**b**) data sending process; (**c**) data receiving process; (**d**) edge prediction and data recovery process.

4. Edge Prediction Algorithm Based on ARMA

4.1. ARMA Model

A time series refers to the results of observing a certain process at a given sampling rate in an equally spaced time period [49]. Environmental monitoring data such as temperature and salinity in the application of underwater three-dimensional monitoring are a form of time series data. The change in the time series has a tendency, periodicity, and seasonality. The change process of the time series can be defined as a set of variables: y_t is dependent on the parameter t. The core of edge prediction is time series prediction, that is, to find out the change law with time of underwater environmental data, such as temperature and salinity, from historical data, and to predict future data.

Considering the computing performance of underwater nodes and the prediction accuracy of underwater monitoring applications, the ARMA model is used as the core model of edge prediction. ARMA is often used for stationary time series prediction [49]. If the monitoring data are a stationary series, the monitoring data can be expressed by the combination of pre-sequence data and prediction error, and the expression is as shown in [49]:

$$y_{t} = \beta_{0} + \beta_{1}y_{t-1} + \dots + \beta_{p}y_{t-p} + \varepsilon_{t} + \alpha_{1}\varepsilon_{t-1} + \dots + \alpha_{q}\varepsilon_{t-q},$$
(1)

where y is the predicted or observed values, and ε indicates the prediction errors; α and β are the coefficients; p and q are the orders, indicating the number of the observed values and prediction errors, respectively. Formula (1) represents time series y_t following the ARMA(p,q) model.

When the monitoring data are non-stationary series, that is to say, the time series contains the characteristics of trend, seasonality and periodicity, the nonstationary sequence can be converted into a stationary sequence through difference processing. The second-order difference expression is as shown in [49]:

$$\nabla^2 \mathbf{y}_t = \nabla (\mathbf{y}_t - \mathbf{y}_{t-1}) = \mathbf{y}_t - 2\mathbf{y}_{t-1} + \mathbf{y}_{t-2}, \tag{2}$$

where ∇ is the difference operator, and ∇^2 is the second-order difference operator. The difference reflects the change between discrete quantities. The first-order difference represents the difference between the current value and the previous value, and the second-order difference represents the difference between the current first-order difference and the previous first-order difference. Formula (2) represents the second-order difference of time series y_t .

 $\nabla^{d}y_{t}$ is the d-order difference of y_{t} After the difference transformation, $\nabla^{d}y_{t}$ becomes a stationary time series, which can be expressed as shown in [49]:

$$\lambda(B)\left(\nabla^{d} y_{t}\right) = \theta(B)\varepsilon_{t},\tag{3}$$

where B is the delay operator, which follows $y_{t-p} = B^p y_t$, $(\forall p \ge 1)$ and $\nabla^k = (1-B)^k$. $\lambda(B) = 1 - \lambda_1 B - \lambda_2 B^2 - \cdots - \lambda_p B^p$ is the autoregressive polynomial, and $\theta(B) = 1 - \theta_1 B - \theta_2 B^2 - \cdots - \theta_q B^q$ is the moving average polynomial. ε_t is a white noise sequence of the zero-mean. Formula (3) is also called the ARIMA(p, d, q) model, which is used to realize edge prediction.

The autocorrelation function (ACF) is often used to determine the order d of the ARIMA model [50–52]. After d-order difference transformation, if the ACF value corresponding to the difference series is close to 0, the time series passes the stationarity test. Smaller values of d represent a better choice.

When selecting the p and q parameters, the Akaike information criterion (AIC) and Bayesian information criterion (BIC) can achieve a better selection effect than using the mean square error (MSE) [53]. AIC is based on the concept of entropy and provides a standard to weigh the complexity of the estimation model and the goodness of the fitting data. AIC is defined in [53] as follows:

$$AIC = -2\log(L) + 2(p + q + k + 1),$$
(4)

where L is the likelihood function of the monitoring data; p and q represent the parameters of the ARMA model, respectively; and k is taken as 1 or 0, indicating whether the error term is considered. When there is a large difference between the two models, the difference is mainly reflected in the likelihood function term. When the likelihood function difference is not significant, the second term of Formula (4), that is, the model complexity, plays a role, so the model with fewer parameters can be selected.

BIC can also realize model selection. BIC is defined in [53] as follows:

$$BIC = AIC + [log(T) - 2](p + q + k + 1),$$
(5)

where T is the number of the monitoring samples. The ARMA (p, q) model with smaller AIC and BIC values not only has a better fitting degree, but also has fewer parameters. The model with fewer parameters can effectively reduce the over-fitting degree of the model.

In this paper, an effective method of automatically traversing the model space is used to select the parameters of the ARIMA(p, d, q) model. The method can select the best d-order, and the p and q parameters according to AIC or BIC at the same time. The method first tries to select several possible parameter models, and then fine tunes the parameters according to the prediction results to quickly match the model parameters of ARIMA(p, d, q) [53].

4.2. EP-ARMA Implementation

The sensor nodes will collect monitoring data sets as $\{collect_i\}$ in the monitoring period T. Additionally, then, based on the ARIMA(p, d, q) model and historical data, the forecast data sets $\{forecast_i\}$ can also be generated, where i is the time point of data collection. When the underwater monitoring data acquisition accuracy is $\{accuracy\}$, the following requirements should be met to ensure the result of edge prediction is suitable for underwater monitoring:

$$\forall i \in \{1, \cdots, T\} | \text{forecast}_i - \text{collect}_i | \le \text{accuracy}, \tag{6}$$

However, due to the randomness and abruptness of time series changes, it is not possible to ensure that all prediction data can fulfil Formula (6). Therefore, a certain amount of collected data should be used as the correction data. When there is a large deviation between the predicted data and the collected data at a certain time point, the collected data at that time point are used to replace the prediction data, and the corrected prediction data set {forecast'_i} and the correction data set {n, i, correct_j} are generated, where n is the number of correction data, i is the correction time point, and *j* is the identification of correction data. By transmitting the prediction parameters and the correction data set, the corrected prediction data set {forecast'_i} that meets the accuracy requirements of Formula (6) can be recovered at the remote end.

The underwater monitoring data acquisition accuracy is dynamically adjusted within the range of {accuracy_{high}, ..., accuracy_{low}}. When the number of correction data sets n is equal to n_{min} , the corrected forecast data set {forecast'_i} can meet the accuracy_{low}. When the number of correction data sets is equal to n_{max} , the corrected forecast data set {forecast'_i} can meet the accuracy_{high}. Therefore, n will change in { n_{min} , ..., n_{max} }. The prediction parameters (p, d, q) and correction time points i are also transmitted when the correction data are transmitted. If the number of correction data sets is taken as n_{min} , the amount of data to be transmitted is still greater than the amount of data to be collected directly, and the algorithm will not be applicable. Assume that p, d, q and i are expressed as integer data, and the collected and correction data are expressed as float data. Then, 1 byte represents the integer data, 4 bytes represent the float data, and when n_{min} cannot meet:

$$n_{\min} < \frac{4 \times T - 3}{5},\tag{7}$$

EP-ARMA will not be suitable for underwater monitoring applications with such accuracy. It is more efficient to directly transmit the collected data or adjust the prediction model. When n_{min} fulfils Formula (7), the number of correction data sets can be adjusted to reduce the amount of data actually transmitted in the underwater network and improve the efficiency of underwater monitoring data transmission. n_{min} is set as the threshold value. Each node needs to ensure that the number of transmission correction nodes is not less than the threshold value.

The value of accuracy_{low} is generally given by the application program, which is an insurmountable threshold for data transmission. The accuracy_{low} is constant in the whole communication process. The value of accuracy_{high} is initially given by the application, but is limited to Formula (7) and the prediction result, and will change dynamically according to the amount of correction data allowed to be transmitted. The value of accuracy_{high} remains constant or decreases in the whole communication process. $\begin{array}{l} \{n_{min}, \cdots, n_{max}\} \text{ will change dynamically with } \left\{ accuracy_{high}, \cdots, accuracy_{low} \right\}, \text{ where } \\ n_{min} \text{ represents the lowest threshold allowed by the service accuracy, and } \\ n_{max} \text{ represents the highest threshold that the edge prediction algorithm can provide. } \\ n_{min}, \cdots, n_{max} \right\} \\ \text{ and } \left\{ accuracy_{high}, \cdots, accuracy_{low} \right\} \\ \text{ play a consistent role in EP-ARMA.} \end{array}$

 $\dot{E}P$ -ARMA, as shown in Algorithm 1, are generally deployed in sensor agents and sink agents. They must find the {p,d,q} parameters with the lowest AIC or BIC value according to the automatic traversal algorithm. Additionally, it generates the prediction data set {forecast_i}. Then, it generates the correction data set {n, i, correct_j} and the threshold value. The edge prediction on the sink agent recovers the corrected prediction data set {forecast'_i}, according to {p,d,q}, {n,i, correct_j}, the threshold value, and the historical data.

Algorithm 1 Edge Prediction Based on ARMA (EP-ARMA)			
Initialize p, d, q range, $\left\{ \text{accuracy}_{\text{high}}, \dots, \text{accuracy}_{\text{low}} \right\}$, forecast step, T			
Complete the monitoring task in period T, and obtain the data of $\{\text{collect}_i\}$			
While (true)			
If (node.type is sensor)			
According to the forecast step, obtain the data _{history} from 'service data' database			
Obtain the ARIMA(p _{best} , d _{best} , q _{best}) by AIC or BIC in p, d, q range			
Obtain the $\{\text{forecast}_i\}$ by ARIMA(p_{best} , d_{best}), $data_{\text{history}}$ and T			
Obtain the $\{n_{min}, \dots, n_{max}\}, \{n, i, correct_j\}$, and threshold by $\{collect_i\}, \{forcast_i\}$,			
$\left\{ accuracy_{high}, \cdots, accuracy_{low} \right\}$ and Formula (7)			
Save the (p_{best} , d_{best} , q_{best}) and $\{n, i, correct_i\}$ in 'service data' database			
Save the 'threshold' in 'transport policy' database			
Wait for the next period T			
End if			
If (node.type is sink && the new data has been received)			
According to the forecast step, get the data _{history} from 'service data' database			
Obtain the (p, d, q), $\{n, i, \text{ correct}_j\}$ and threshold from the received data			
Obtain the {forecast'_i} by ARIMA (p, d, q) and {n, i, $correct_j$ }			
Save the {forecast'_i} in 'service data' database			
End if			
end while			

5. Adaptive Data Transmission Algorithm Based on Reinforcement Learning

5.1. Reinforcement Learning Model

Reinforcement learning is a kind of learning that aims to obtain the best mapping strategy from environmental state to action. The best strategy can maximize the reward, which is the evaluation of the movement quality. Due to the limited information provided by the external environment, reinforcement learning can improve the ability to adapt to the dynamic environment by continuously learning and acquiring feedback knowledge in the process of action evaluation. Reinforcement learning algorithms are often described by Markov decision processes (MDPs). The process of reinforcement learning adopts the five tuples of (S, A, P, R, γ), where S stands for environment status, A stands for action, P stands for transition probability, R stands for reward, and γ represents the discount rate. Agents, according to the environment status S, cumulative reward R, and discount rate γ , determine the transition probability P, and generate action A, then switch to the new state S *t* [48]. In underwater wireless sensor networks, the process of data transmission can also be regarded as Markov decision processes (MDPs). Reinforcement learning is used to select the best relay node and meet the required data transmission accuracy, as shown in Figure 4.



Figure 4. Framework of adaptive data transmission model based on reinforcement learning.

Assuming that the nodes of the underwater wireless sensor network can be expressed as follows:

$$N = \{n_1, n_2, n_3, \cdots, n_m\},$$
(8)

where *n* represents the node and *m* is the number of nodes, each node in the network can obtain its own location information. Therefore, a virtual pipeline can be built on the vector between the data source node and the target sink node, as shown in Figure 5. The virtual pipe radius of the node i is set to:

$$R^{i}_{pipe} = -\frac{R - R^{i}_{ini}}{E_{ini}} \times \overline{E^{i}} + R , \qquad (9)$$

where E_{ini} is the initial energy of the node, E^i is the average residual energy of the neighbor nodes, R is the signal transmission distance of the node, and R^i_{ini} is the initial radius of the pipeline. When the neighbor nodes have more residual energy, they can maintain a small pipeline radius, allowing the data to acquire the shortest path. When the neighbor nodes have less residual energy, they can enlarge the pipeline radius of transmission, allowing more neighbor nodes to participate in packet forwarding.

The candidate relay node set $N_{relay}(i)$ of the current node n_i , can be expressed as follows:

$$N_{relay}(i) = \left\{ n_j \in N_i(t) \subseteq \left\{ N | dep(n_j) - dep(n_i) \le 0 \right\} \cap neighbors(n_i) \cap pipe(n_i) \right\},$$
(10)

At time *t*, the node set of candidate relay nodes for n_i is $N_i(t)$. The node set of neighbors overridden by the one-hop transmission of the node n'_i s signal is neighbors (n_i) . The node set in the n'_i s virtual pipe is pipe (n_i) . The node set with a shallower depth than the current node n_i is $\{N | dep(n_j) - dep(n_i) \le 0\}$. The relay node selection needs to be within the range defined by Formula (10).

Definition 1. *At time t, if the packet to be sent is at node* n_i *, then the current environment state S can be defined as follows:*

$$S = \bigcup_{acc} \left(\{n_i\} \cup N_{relay}(i, acc) \right), \tag{11}$$

where 'acc' denotes that the monitoring data are located in the variable accuracy range of the node, which is determined by $\left\{ accuracy_{high}, \cdots, accuracy_{low} \right\}$ or $\{n_{min}, \cdots, n_{max}\}$. The environment includes not only all possible relay nodes, but also all possible data accuracy on different nodes.



Figure 5. Schematic diagram of virtual pipeline. The candidate relay nodes in the virtual pipeline are represented in red, and the color of the sensor nodes represents the type of collected data.

Definition 2. At time t, action A can be defined as follows:

$$A = \left\{ a_{j,acc} | (n_j \in S, acc \in \{n_{\min}, \cdots, n_{\max}\}) \right\},$$
(12)

If the relay node completes data forwarding, it will be rewarded. If the relay node is closer to the water surface and has more residual energy, and the delay of transmission channel is smaller, it will be more conducive to the efficient transmission of data to the sink node, and it will be given additional rewards. If it adopts higher transmission accuracy and transmits more correction data, it will also receive additional rewards.

Definition 3. At time t, if the packet is at node n_i , and n_i selects n_j as a relay node, and sends the packet with the accuracy 'acc', the reward function is as follows:

$$R_{n_{l}n_{j,acc}}^{a_{j,acc}} = -R_{0} - \left[\phi_{e} \times co(e) + \phi_{t} \times co(t) + \phi_{l} \times co(l)\right],$$
(13)

Part I: the fixed cost R_0 will be accumulated with each data forwarding. The existence of R_0 can help the agent select a route with fewer hops. However, the shortest route is not always the best route. It also needs to comprehensively consider the factors such as the nodes, channels, and data transmission accuracy.

Part II: co(e) is the cost for node energy. The existence of co(e) can help the agent select the node having larger residual energy as the relay node, which is more conducive to the continuation of the network lifecycle. co(e) can be expressed as follows:

$$co(e) = 1 - \frac{E_{res}^{j}}{\sum_{k \in N_{relay}(i)} E_{res}^{k}}, \qquad (14)$$

where E_{res}^{j} represents the residual energy of the next hop node, $\sum_{k \in N_{relay}(i)} E_{res}^{k}$ represents the total remaining energy of the candidate node set, and φ_{e} is the sensitivity coefficient of residual energy cost of neighbor nodes.

Part III: co(t) is the cost for the delay of the transmission channel. The existence of co(t) can help the agent select the node having smaller transmission delay as the relay node. This means that the channel between the current node and the relay node is stable and reliable, has a low bit error rate and has less congestion. co(t) can be expressed as follows:

$$co(t) = 1 - \beta_1 \frac{1}{t_{total}^{i \to j} + 1} - \beta_2 \frac{1}{t_{receive}^{j} + 1}$$
, (15)

where $t_{total}^{i \rightarrow j}$ is the average of total delay from node n_i to node n_j . The total delay refers to the time from n_i sending the data until n_i hears n_j forwarding the same data, which includes the sender delay in n_i , the propagation delay from n_i to n_j , the processing and waiting delay in n_j , and the propagation delay from n_i to n_i , the processing and waiting delay in n_j , and the propagation delay from n_i to n_i . transmission failure. It is generally defined according to the transmission conditions and business requirements. $\overline{t_{receive}}^j$ is the average transmission delay from n_j . ϕ_t is the sensitivity coefficient of the transmission channel delay cost. β_1 and β_2 are used to adjust the weight of the two delay parameters, respectively. $\overline{t_{total}}^{i \rightarrow j}$ is the actual delay of packet transmission. $\overline{t_{receive}}^j$ can help nodes to optimize transmission policies without forwarding the packets.

Part IV: co(1) *is the cost related to data transmission accuracy. The data transmission accuracy is determined by the number of correction data transmitted.* co(1) *can be expressed as follows:*

$$co(l) = \frac{1}{l_{correct} + 1} , \qquad (16)$$

 $l_{correct}$ indicates the number of correction data sent. The value range of $l_{correct}$ is [threshold, \cdots , $l_{correct}^{max}$]. $l_{correct}^{max}$ is the number of correction data received or generated by the node n_i . φ_1 is the sensitivity coefficient of the correction accuracy cost. Selecting higher φ_1 can reduce the accuracy of the monitoring data. However, the greater length of the transmitted data can reduce the packet-sending success rate, and increase the transmission delay. Although selecting lower φ_1 will reduce the accuracy of data transmission, it can improve the reliability of data transmission. Therefore, selecting the appropriate φ_1 will realize the win–win of transmission efficiency and transmission accuracy.

Definition 4. At time t, if the current packet is in node n_i , the transition probability of node n_i to node n_j the accuracy 'acc' is defined as follows:

$$P_{n_{i}n_{j,acc}}^{a_{j,acc}} = \frac{R_{n_{i}n_{j,acc}}^{a_{j,acc}}}{\sum_{n_{k,acc'} \in S} R_{n_{i}n_{k,acc'}}^{a_{k}}},$$
(17)

In order to reflect the impact of the future status on the current status, the overall reward at time t is defined as shown in [5]:

$$R_{t} = r_{t} + \gamma r_{t+1} + \gamma^{2} r_{t+2} + \dots = \sum_{j=0}^{\infty} \gamma^{j} r_{t+j} , \qquad (18)$$

when $\gamma = 0$, the system will only consider the reward r of the current action, not the future status. When $\gamma \to 1$, the system will consider both the current and future action rewards r.

According to Q-learning, the state action function under policy π is defined as shown in [5]:

$$Q^{\pi}(s,a) = E_{\pi}\{R_t | s_t = s, a_t = a\},$$
(19)

Assuming that a 'and s ' are the next action and state, the optimal solution $Q^*(s, a)$ in the state of $Q^{\pi}(s, a)$ can be expressed as an iterative equation, as shown in [5]:

$$Q^{*}(s,a) = r_{t} + \gamma \sum_{s \prime \in S} P^{a}_{s s \prime} \left\{ \max_{a \prime} Q^{*}(s \prime, a \prime) \right\},$$
(20)

According to the state action function, the node with sufficient residual energy and smaller transmission channel delay in the virtual pipeline will be selected as the relay node. The packets will be packaged according to the appropriate transmission accuracy meeting the threshold requirements and sent by the current node.

At time t, the value function V will select the state action function that can obtain the maximum benefit, which is defined as shown in [5]:

$$V_t^*(s) = \max Q^*(s, a)$$
, (21)

At the initial stage of reinforcement learning, the initial V value of each agent is 0. With the accumulation of learning, the V value will be continuously updated according to Formulas (20) and (21) and gradually converge. A good data forwarding strategy will finally emerge.

5.2. RL-ADTA Implementation

RL-ADTA is implemented, as shown in Algorithm 2. The agents in underwater wireless sensor networks can obtain the status information of neighbor nodes and transmission channels by monitoring the packet information transmitted in the network. When a new packet is generated or received, the Q value is calculated according to the environment state of the current node where the packet is located, and the best policy is selected to send the packet. Once the packet is sent, the agent will record whether the packet is successfully sent, and update the agent state.

Algorithm 2 Adaptive Data Transmission Algorithm Based on Reinforcement Learning (RL-ADTA)			
Initialize the positions of the nodes			
While (true)			
If (the new data to send)			
While (times _{send} $<$ times _{max})			
Create the virtual routing pipe using Formula (9)			
Obtain the status of neighbors, channels, and the current node from the environment			
status base			
Obtain the V-value, threshold, and super-parameters from the transport policy base			
Update the S and A using Formulas (10)–(12)			
Calculate Q function using Formulas (13)–(20)			
Update the V-value with the max Q value using Formula (21)			
Determine the number of correction data sent n _{new} by the max Q value			
Determine the relay node using the max Q value			
Form the packet by (p, d, q), $\{n_{new}, i, correct_i\}$ and V-value			
Forward the packet to the relay node			
If (It is detected that the packet has been forwarded)			
Break			
Else			
times _{send} ++			
End If			
End While			
End if			
End While			

5.3. RL-ADTA Packet Design

The data transmission algorithm between the nodes is finally realized by transferring the packets containing the control and application information. RL-ADTA does not generate special requirements for the lower layer protocol, nor does it design data packets separately to transfer status or control information between nodes. The packet defined by RL-ADTA contains both the service information and the node status information. The underwater network nodes update the environment status by monitoring the packets sent by the neighbor nodes. When the relay node is the current node, the optimal data forwarding strategy is obtained according to RL-ADTA, and the packet is repackaged and forwarded to the next node. The packet structure of RL-ADTA is shown in Figure 6. The data package contains two parts: the fields in the 'data head' are related to underwater communication, while the fields in the 'data content' are related to monitoring applications.



Figure 6. RL-ADTA packet structure. Figure (**a**) shows the composition of RL-ADTA package. Figure (**b**) shows the constituent fields of the 'data content'.

The underwater monitoring application forms the monitoring data into data packets during every period T. The 'data ID' is the ID of the current period data, and the 'destination address' is the ID of the destination sink node. The 'node ID', 'V value', 'residual energy' of the current node, the 'timestamp' of sending data, and the 'next forward' representing the relay node ID are all included in the data header. The fields 'task type ' and 'threshold' in the data content are determined by the type of monitoring data collected. Additionally, (p, d, q), correction point', {correct₁, \cdots , correct_{num}} and 'num' are the edge prediction parameters, correction time-point, correction data, and their quantities, respectively.

6. Result and Discuss

6.1. Experimental Environment and Data

The experiment was based on the application of underwater temperature monitoring. The average depth of the ocean is within 3000 m. At the same water depth, the seawater temperature in the range of 5000 m \times 5000 m is generally considered to be similar. Therefore, in the experiment, the range of underwater network was set to 5000 m \times 5000 m \times 2500 m. The transmission medium adopts an acoustic channel. The transmission speed of sound in water is affected by depth, temperature, and salinity. In the experiment, the sound velocity in seawater was set to 1500 m/s according to the common setting. The distance of sound propagation is related to its frequency. The communication distance of medium distance underwater acoustic communication is generally 1-10 km, and the bandwidth is in the order of 10 kHz. In the experiment, the frequency of sound was set to 10 kHz, and the communication distance and sensor distance were set to 1000 m, while the initial width of the virtual pipe was set to half of the communication distance. The adaptability of the algorithm should be verified by changing the density of different nodes. Therefore, in the experiment, 100, 200, or 300 nodes were selected to participate in the experiment. The initial energy of each node is 1000 J, the transmission power of the node is 10 W, the receiving power of the node is 3 W, the calculation energy power of the node is 48 mW, and the idle power of the node is 30 mW. The experimental data are derived from the temperature data observed 400 m underwater at the KEO station in the database of the National Oceanic and Atmospheric Administration (NOAA). In the data transmission experiment, the temperature data collected every 6 h are formed into data packets for transmission, the size of the original data packets is 50 bytes, and the transmission rate of the application is 1 kbps. In order to shorten the simulation process, the sending rate of the packet is 0.1 packet/s, and the simulation time is 2000 s. The temperature accuracy of 0.01, 0.1, and 0.2 °C was selected as the lowest accuracy allowed for application. All parameter settings of the experiment are displayed in Table 2.

Table 2. Experiment settings.

Name	Value
Underwater network	5000 m \times 5000 m \times 2500 m
Transfer speed of sound	1500 m/s
Frequency of sound	10 kHz
Communication range of the node	1000 m
Sensor range of the node	1000 m
Initial width of the virtual pipe	500 m
Number of nodes	100, 200, 300
Initial energy of the node	1000 J
Transmission power of the node	10 W
Receiving power of the node	3 W
Calculation energy power of the node	48 mW
Idle power of the node	30 mW
Source of underwater temperature data	NOAA-KEO (-400 m)
Monitoring data transmission interval	6 h
Origin data packet size	50 Bytes
Generation rate of the packet	0.1 packet/s
Transmission rate of the application	1 kbps
Simulation time	2000 s
Lowest temperature accuracy	0.01, 0.1, 0.2 °C
$R_0, \varphi_e, \varphi_t, \varphi_l, \gamma, \beta_1, \beta_2$	-1, 0.5, 0.5, 0.1, 0.1, 0.7, 0.3

The simulation environment was built based on Python 3.9, and the EP-ARMA and RL-ADTA algorithms were also implemented based on Python 3.9. The simulation environment comprised a computer having an i7-9700k CPU with a dominant frequency of 3.6 GHz, 64 GB memory, and GT1080 GPU.

The data are the underwater temperature data at 400 m below the KEO station from 12:00:00 on 16 June 2004 to 11:00:00 on 23 November 2004, i.e., a total of 3840 periods, as shown in Figure 7.



Figure 7. Temperature data at 400 m underwater of KEO station.

In order to verify the stability of the experimental data, the data distribution is formed after the first-order transformation, as shown in Figure 8.



Figure 8. First-order transformation data of temperature data of KEO station at 400 m underwater. After the first-order difference, the change in temperature data tends to be stable.

6.2. Data Prediction Performance Analysis

6.2.1. Prediction Accuracy Analysis

The data of KEO station are collected every hour. The 40-day (960 h) temperature data of four periods are used to predict the data of the next day (24 h) through the ARMA and EP-ARMA algorithms, respectively. The ARMA algorithm adopts the BIC value to automatically select the ARMA parameters. The lowest temperature accuracy of the EP-ARMA algorithm is 0.1 °C. The prediction performance analysis data are shown in Figure 9.

The left parts, (a), (c), (e) and (g), of Figure 9 show that the ARMA model based on the automatic optimization of parameters through BIC can indeed accurately use the historical data of the first 960 periods to predict the change trend of the temperature data in the next 24 h, but there are still some errors at each time point, especially at some key points with drastic changes. Although the EP-ARMA (0.1) model only corrects the data whose error between the predicted value and the monitored value exceeds 0.1 °C based on the prediction results of the ARMA model, it is shown by (b), (d), (f) and (h) on the right side of Figure 9 that the corrected data can accurately reflect the temperature data of the next 24 h in the four periods, and the key data with sharp changes in the period can also be extracted. Even in the face of temperature data in different periods, EP-ARMA can achieve more accurate prediction, especially to ensure that abnormal temperature changes are not omitted. Therefore, EP-ARMA can meet the requirements of data transmission accuracy for underwater monitoring applications, and this prediction ability is robust.

6.2.2. Prediction Parameter Analysis

In time series analysis and prediction, the mean absolute error (MAE), mean square error (MSE), and root mean square error (RMSE) can calculate the deviation between the predicted value and the original value. The larger the deviation, the larger the MAE, MSE, and RMSE; the smaller the deviation, the smaller the MAE, MSE, and RMSE. Therefore, the MAE, MSE, and RMSE are often used to evaluate the performance of prediction algorithms [54]. Among these, the MAE is defined as shown in [54]:

MAE =
$$\frac{1}{N} \sum_{x=1}^{N} |p_x - \hat{p}_x|$$
, (22)

The MSE is defined as shown in [54]:

MSE =
$$\frac{1}{N} \sum_{x=1}^{N} (p_x - \hat{p}_x)^2$$
, (23)

The RMSE is defined as shown in [54]:

RMSE =
$$\sqrt{\frac{1}{N} \sum_{x=1}^{N} (p_x - \hat{p}_x)^2}$$
, (24)

where, the monitoring data are denoted by p_x , the forecast data are denoted by \hat{p}_x , the length of the data series is denoted by N, and x denotes the label of the data series.

Using the 40-day (960 h) temperature data in the four periods, ARMA models with different (p, d, q) parameters were selected to predict the data of the next day (24 h). The MAE, MSE, and RMSE curves calculated using the predicted data and the original monitoring data are shown in Figure 10.



Figure 9. Comparison of ARMA and EP-ARMA algorithm prediction data and original monitoring data in different periods. Figures (**a**,**c**,**e**,**g**) in the left part reflect the comparison between the original monitoring data and ARMA prediction data in the periods (0, 960), (960, 1920), (1920, 2880), and (2880, 3840), and in the next 24 h period. Figures (**b**,**d**,**f**,**h**) in the right part reflect the comparison between the original monitoring data and EP-ARMA (0.1) correction data in the periods (0, 960), (960, 1920), (1920, 2880), and (2880, 3840), and in the next 24 h period.



Figure 10. Prediction performance analysis of different parameters selected using ARMA in different periods. Based on the temperature data of (0, 960), (960, 1920), (1920, 2880), (2880, 3840) periods, ARMA models with different (p, d, q) parameters are used to predict the temperature data of the next 24 h period; calculate the MAE, MSE and RMSE formed by the predicted values and the original monitoring values; and draw the analysis curves (**a**–**d**).

Figure 10 shows that the curves of MAE, MSE and RMSE vary sharply with different (p, d, q) configurations. The curves in different periods show that the selection of different (p, d, q) values has an important impact on the prediction performance of EP-ARMA. The (p, d, q) configuration with the highest prediction accuracy is not completely consistent in different periods. Therefore, high-precision EP-ARMA prediction cannot be achieved with a fixed (p, d, q) configuration. It is necessary to find a new (p, d, q) configuration through the automatic parameter-matching method after a period of time, to ensure the stability of the predictive performance.

Figure 10 also shows that, due to the stability of the underwater temperature data, it is not necessary to change the (p, d, q) every period. The parameters in multiple consecutive periods do not change to a large extent. Therefore, it is only necessary to dynamically adjust the parameter of (p, d, q) when the prediction performance is greatly reduced. In the case of poor transmission conditions or stable changes in monitoring data, the data updating frequency can be reduced or even stopped. The subsequent data can be generated using EP-ARMA and historical data based on the (p, d, q) in the previous period, which can greatly improve the data transmission efficiency of the EP-ADTA.

6.2.3. Forecast Model Analysis

Exponential smoothing (ES) is a classical time series prediction method [55]. The prediction generated by the exponential smoothing algorithm is the weighted average of the past observations. As the number of observations grows, the weighted values decay exponentially. This is suitable for short- and medium-term forecasting and has good accuracy. The prediction accuracy of the ES model depends on the smoothing factor

 α , which is set between 0.1 and 0.9. The smoothing factor α can be acquired using the trial-and-error method. In this experiment, the value of α is 0.22.

GRU is a variant of the RNN network and a type of long-term memory network that can effectively capture the nonlinear relationship between sequence data [56,57]. GRU adds a gating unit to the standard RNN model. The gating unit can control the flow status of information in different time steps in the network. GRU is a simplified version of the LSTM network, but the number of parameters used is still huge. The GRU structure used in the experiment adopts the structure of two layers of GRU and one layer of full connection, having 74,661 parameters.

The ES, ARMA, GRU, and EP-ARMA (0.1) models were used to predict the underwater temperature data based on three different training and prediction periods. The RMSE representing the performance of the four methods is shown in Figure 11.



Figure 11. Comparison of RMSE of prediction performance of ES, ARMA, GRU, and EP-ARMA algorithms. '168-4' means that the training data cover 168 periods, and the prediction data cover 4 periods. '960-24' means that the training data cover 960 periods and the prediction data cover 24 periods. '1920-48' means that the training data cover 1920 periods and the prediction data cover 48 periods.

Figure 11 shows that ES, ARMA, GRU, and EP-ARMA have similar prediction effects when forecasting with '168-4' and '960-24'. ARMA, GRU and EP-ARMA have better prediction effects when forecasting with '1920-48'. Compared with GRU, ARMA and EP-ARMA have fewer parameters and smaller computational complexity, which is very important for resource-constrained underwater communication nodes. As the EP-ARMA model corrects the data, it gains a large return in accuracy; therefore, its RMSE is the lowest under the '168-4', '960-24', and '1920-48' combinations. When selecting the prediction model, we should comprehensively consider the prediction performance and computing performance. Among the above four models, the EP-ARMA model is the best.

6.3. Transmission Efficiency Analysis

6.3.1. Transmission Accuracy Analysis

According to the research of this paper, in the field of underwater wireless sensor networks, there is no algorithm that can dynamically adjust the accuracy of data transmission when selecting routes and sending data. Therefore, the existing state-of-the-art algorithms were not compared with the RL-ADTA in the analysis of the transmission accuracy. The deviation degree between the data collected and sent by the sensor node and the data received and recovered by the sink node is used to illustrate the performance of the RL-ADTA in controlling the accuracy of the transmitted data.

RL-ADTA (0.1, 0.2, 0.01) was used to carry out underwater temperature monitoring and data transmission experiments based on 20 different periods. The accuracy_{low} of RL-ADTA (0.1, 0.2, 0.01) determines the actual minimum amount of data to be transmitted, that is n_{min} (0.1, 0.2, 0.01). The n_{max} of RL-ADTA (0.1, 0.2, 0.01) are determined using Formula (7). When the monitoring data transmission interval is 6, the n_{max} (0.1, 0.2, 0.01) are all set to be 4. The number of transmission correction data initially set by RL-ADTA (0.1, 0.2, 0.01) is 4, which represents the accuracy_{high} of RL-ADTA (0.1, 0.2, 0.01). During transmission, the accuracy of RL-ADTA (0.1, 0.2, 0.01) remains unchanged or gradually decreases according to the transmission environment, but the deviation is not lower than the accuracy_{low} (0.1, 0.2, 0.01).

After several rounds of iteration and training, the RL-ADTA (0.1, 0.2, 0.01) model converges. The data of 20 periods are sent into the underwater network transmission model based on RL-ADTA (0.1, 0.2, 0.01). The ratio between the data actually received by the sink node and the data that need to be sent by the sensor node, and the RMSE between the recovered data in sink node and the collected data in sensor node, are calculated and shown in Figure 12.



Figure 12. Analysis of transmission efficiency and accuracy of RL-ADTA (0.1, 0.2, 0.01). The colors of lines and histograms represent the type of algorithm. The ratio between the data actually received by the sink node and the data that need to be sent by the sensor node in each period is shown in the line chart. The lower the ratio, the smaller the amount of correction data received. The RMSE between the data recovered by the sink node according to the prediction parameters and correction data, and the data collected by the sensor node in each period, is shown in the histogram. The smaller the RMSE, the higher the accuracy of the underwater monitoring application.

Figure 12 shows that, as the accuracy_{low} decreases, the RMSE of each period gradually decreases, but the number of data transmitted gradually increases. When the accuracy_{low} is 0.2, the mean value of the RMSE in 20 periods is 0.0903, the mean value of the received/sent ratio is 0.2583; and there is no need to send correction data in periods 7, 15, 16, 19, and 20. When the accuracy_{low} is 0.1, the mean value of the RMSE in 20 periods is 0.0473, the mean value of the received/sent ratio is 0.5292, and the number of sent correction data fluctuates greatly. When the selected threshold is 0.01, the mean value of the RMSE in 20 periods is 0.0286, the mean value of the received/sent ratio is 0.9021, and the number of the correction data actually received by the sink node is close to the number of the collected data that needs to be sent by sensor node. According to the data analysis—although the initial accuracy of RL-ADTA (0.1, 0.2, 0.01) is set to the same accuracy_{high}—due to the difference in accuracy_{low} (0.1, 0.2, 0.01), and the fact that the parameter setting pays more attention to the packet-sending success rate and transmission efficiency, through multiple episodes of training, the transmission accuracy of RL-ADTA (0.1, 0.2, 0.01) converges to the accuracy_{low} (0.1, 0.2, 0.01), respectively. The simulation shows that RL-ADTA (0.1, 0.2, 0.01) can meet the requirements of different levels of monitoring application data transmission accuracy. The priority of efficiency and accuracy of underwater network data transmission can be adjusted through parameter configuration.

6.3.2. Transmission Delay Analysis

RCAR [5] is a state-of-the-art reinforcement learning routing algorithm for underwater wireless sensor networks with the goal of avoiding congestion. RL-ADTA (0.1, 0.2, 0.01) and RCAR algorithms conduct a performance comparison in terms of three aspects: transmission delay, energy consumption, and packet-sending success rate.

The number of nodes in the simulation environment was set to 100, 200, and 300, respectively. RL-ADTA (0.1, 0.2, 0.01) and RCAR were used for the comparison of transmission delay. The average end-to-end delays of data packets from generation and transmission to reception are shown in Figure 13.



Figure 13. RCAR and RL-ADTA (0.1, 0.2, 0.01) delay comparison. The end-to-end delay of RCAR and RL-ADTA includes transmission delay, propagation delay, and processing delay.

Figure 13 shows that the trend in the delay change for the four algorithms is that the end-to-end delay decreases with an increase in the number of nodes. As the number of nodes increases, more and better routes are generated, and the delay is further reduced. As the amount of transmitted data in the RL-ADTA is reduced due to edge prediction, even considering the certain calculation delay, the transmission delay of RL-ADTA (0.1, 0.2) is still better than that of RCAR. RL-ADTA (0.01) does not compress a sufficient amount of the data sent, and adds a fixed calculation delay, which makes the delay larger than that of RCAR. Compared with RCAR, when the number of nodes is 100, RL-ADTA (0.1) reduces the delay by 9.58%; and when the number of nodes is 300, RL-ADTA (0.1) reduces the delay by 0.21%.

In the experiment, in each period when data is sent, it is mandatory to undertake an automatic traversal process to select the best (p, d, q), and then carry out edge prediction. According to the results of 'prediction parameter analysis', (p, d, q) changes slowly. The previous set of (p, d, q) parameters can be first used for edge prediction; it is then decided whether to update (p, d, q). By reducing the update frequency of (p, d, q), the end-to-end delay of RL-ADTA can be further compressed.

6.3.3. Energy Consumption Analysis

RL-ADTA (0.1, 0.2, 0.01) and RCAR are used for the comparison of energy consumption. The total energy consumption of all the nodes in the underwater network during the simulation are considered. The average energy consumption is shown in Figure 14.





Figure 14 shows that the trend in energy consumption of the four algorithms is that, as the number of nodes increases, the energy consumption also increases synchronously. As the node density increases, more nodes participate in the process of data forwarding, causing the energy consumption to gradually increase. The RL-ADTA uses the design of the 'virtual pipeline', like that of RCAR, which reduces the training space of reinforcement learning, shortens the convergence time, and reduces the energy consumption. In the selection of relay nodes, it pays more attention to the location of relay nodes and the delay

of the transmission channel, making the selection range of relay nodes more accurate and further reducing the energy consumption. RL-ADTA makes use of all kinds of information sent by the neighbor nodes to form the state information of the underwater network, instead of using ACK packets; thus, it further reduces the communication energy consumption. The most important application is edge prediction, which greatly reduces the amount of data transmitted in RL-ADTA. Therefore, the energy consumption of RL-ADTA is greatly reduced compared with that of RCAR. Compared with RCAR, when the number of nodes is 100, RL-ADTA (0.1) reduces the energy consumption by 44.4%; when the number of nodes is 200, RL-ADTA (0.1) reduces the energy consumption by 44.7%; and when the number of nodes is 300, RL-ADTA (0.1) reduces the energy consumption by 45.2%.

Due to the difference in the accuracy_{*low*} setting of the RL-ADTA algorithm, RL-ADTA (0.01) sends the most data, RL-ADTA (0.1) takes second place, and RL-ADTA (0.2) sends the least data. This is why RL-ADTA (0.2) consumes the least energy, followed by RL-ADTA (0.1) and RL-ADTA (0.01). The RL-ADTA consumes less computational energy, but saves more communication energy.

6.3.4. Packet-Sending Success Rate Analysis

RL-ADTA (0.1, 0.2, 0.01) and RCAR are used for the comparison of the packet-sending success rate. During the simulation, the RL-ADTA and RCAR sent the same batch of data, and the actual amount of data sent by the four algorithms was required to be the same. The packet-sending success rate is shown in Figure 15.



Figure 15. Comparison of packet-sending success rate of RCAR and RL-ADTA (0.1, 0.2, 0.01). Packet -sending success rate refers to the ratio of the data received and recovered by the sink node to the data sent and collected by the sensor node.

Figure 15 shows that the trend in the change in the packet-sending success rate of the four algorithms is that, with the increase in the number of nodes, the packet-sending success rate also increases synchronously. As the node density increases, more nodes participate in the data transmission, which improves the packet-sending success rate. In the experiment, the RL-ADTA has a larger data margin to retransmit data because the actual transmission data are compressed. Under the same conditions, the packet-sending success

rate of RL-ADTA is greatly improved compared with that of RCAR. When the accuracy_{low} is set to 0.2, the packet-sending success rate of RL-ADTA is close to 100%. Compared with RCAR when the number of podes is 100, RL-ADTA (0.1) improves the packet-sending

RCAR, when the number of nodes is 100, RL-ADTA (0.1) improves the packet-sending success rate by 32.2%; when the number of nodes is 200, RL-ADTA (0.1) improves the packet-sending success rate by 43.4%; and when the number of nodes is 300, RL-ADTA (0.1) improves packet-sending success rate by 20.7%.

The amount of data actually transferred in underwater wireless sensor networks directly affects the efficiency of data transmission. Too much data transmission will increase the delay and energy consumption, and reduce the packet-sending success rate. However, too little data transmission will affect the accuracy of underwater monitoring applications. The location and residual energy of the relay nodes and the delay of the transmission channels also affect data transmission efficiency. Good routes will enable the packets to reach the target node with less forwarding nodes and less delay, balancing the energy consumption and extending the underwater network lifecycle. RL-ADTA can autonomously adjust the data transmission accuracy and the selection of routes to adapt to the changes in the underwater network. Through the comparative experiments, it is not difficult to find that the overall performance of RL-ADTA is significantly improved compared with RCAR. The better routing selection and more flexible control of data transmission accuracy makes RL-ADTA produce more computing delay and energy consumption, but the total delay and energy consumption are still less than those of RCAR, and the packet-sending success rate is higher. RL-ADTA obtains greater benefits at a low cost. The exchange of computing resources and communication resources is generally worthwhile. Moreover, with the improvement in of the prediction algorithm and computing power, the advantages will become increasingly obvious. Therefore, the data transmission efficiency of the RL-ADTA will be greater in future applications.

7. Conclusions

An adaptive data transmission algorithm for underwater wireless sensor networks, EP-ADTA, is proposed in this paper. The end-edge-cloud architecture is used to define the underwater wireless sensor networks, and the agent is used to define the network nodes. All the monitoring data that enter the edge network are compressed and preprocessed using the edge prediction algorithm, EP-ARMA, and only the prediction parameters and correction data are transmitted in the network transmission process. The selection of relay nodes and the adjustment of transmission accuracy are controlled by the reinforcement learning algorithm, RL-ADTA. The algorithm comprehensively considers the location, energy, and channel delay of the transmission node, and sends the data packets with appropriate accuracy to the best relay node. By comprehensively considering the functions and requirements of all layers and using the method of edge computing to compress the demand of actual node data communication, the algorithm realizes more efficient and reliable communication in underwater wireless sensor networks. Simulation experiments show that EP-ADTA can meet the needs of underwater monitoring applications having different levels of data acquisition accuracy, and can effectively compress the actual amount of data transmitted. Compared with the state-of-the-art algorithms, EP-ADTA (0.1) reduces transmission delay by 9%, reduces energy consumption by 44%, and improves the packetsending success rate by 35%. With the development of underwater network computing power, the underwater nodes will gain more edge computing functions, which will achieve more efficient and reliable data communication.

Author Contributions: Conceptualization, B.W., K.B. and H.L.; methodology, B.W. and K.B.; software, B.W.; validation, B.W. and M.Z.; resources, B.W.; data curation, B.W.; writing—original draft preparation, B.W. and F.Z.; writing—review and editing, B.W.; funding acquisition, M.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by National Natural Science Foundation of China, grant number 52071153.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Fattah, S.; Gani, A.; Ahmedy, I.; Idris, M.Y.I.; Hashem, I.A.T. A Survey on Underwater Wireless Sensor Networks: Requirements, Taxonomy, Recent Advances, and Open Research Challenges. *Sensors* 2020, *20*, 5393. [CrossRef]
- Qiu, T.; Zhao, Z.; Zhang, T.; Chen, C.; Chen, C.L.P. Underwater Internet of Things in Smart Ocean: System Architecture and Open Issues. *IEEE Trans. Ind. Inform.* 2020, 16, 4297–4307. [CrossRef]
- Haque, K.F.; Kabir, K.H.; Abdelgawad, A. Advancement of Routing Protocols and Applications of Underwater Wireless Sensor Network (UWSN)—A Survey. J. Sens. Actuar. Netw. 2020, 9, 19. [CrossRef]
- 4. Jiang, J.; Han, G.; Shu, L.; Chan, S.; Wang, K. A Trust Model Based on Cloud Theory in Underwater Acoustic Sensor Networks. *IEEE Trans. Ind. Inform.* 2017, 13, 342–350. [CrossRef]
- Jin, Z.; Zhao, Q.; Su, Y. RCAR: A Reinforcement-Learning-Based Routing Protocol for Congestion-Avoided Underwater Acoustic Sensor Networks. *IEEE Sens. J.* 2019, 19, 10881–10891. [CrossRef]
- Basagni, S.; Petrioli, C.; Petroccia, R.; Spaccini, D. CARP: A Channel-Aware Routing Protocol for Underwater Acoustic Wireless Networks. *Ad Hoc Netw.* 2015, 34, 92–104. [CrossRef]
- Lu, Y.; He, R.; Chen, X.; Lin, B.; Yu, C. Energy-Efficient Depth-Based Opportunistic Routing with Q-Learning for Underwater Wireless Sensor Networks. Sensors 2020, 20, 1025. [CrossRef]
- Datta, A.; Dasgupta, M. Energy Efficient Layered Cluster Head Rotation Based Routing Protocol for Underwater Wireless Sensor Networks. Wirel. Pers. Commun. 2022, 1–18. [CrossRef]
- 9. Chenthil, T.R.; Jayarin, P.J. An Energy-Aware Multilayer Clustering-Based Butterfly Optimization Routing for Underwater Wireless Sensor Networks. *Wirel. Pers. Commun.* 2022, 122, 3105–3125. [CrossRef]
- 10. Liu, J.; Yu, M.; Wang, X.; Liu, Y.; Wei, X.; Cui, J. RECRP: An Underwater Reliable Energy-Efficient Cross-Layer Routing Protocol. Sensors 2018, 18, 4148. [CrossRef]
- 11. Faheem, M.; Tuna, G.; Gungor, V.C. QERP: Quality-of-Service (QoS) Aware Evolutionary Routing Protocol for Underwater Wireless Sensor Networks. *IEEE Syst. J.* 2018, 12, 2066–2073. [CrossRef]
- 12. Bai, Q.; Jin, C. A K-Means and Ant Colony Optimization-Based Routing in Underwater Sensor Networks. *Mob. Inf. Syst.* 2022, 2022, 4465339. [CrossRef]
- 13. Barthwal, N.; Verma, S.K. An Optimized Routing Algorithm for Enhancing Scalability of Wireless Sensor Network. *Wirel. Pers. Commun.* **2021**, *117*, 2359–2382. [CrossRef]
- 14. Rao, P.C.S.; Lalwani, P.; Banka, H.; Rao, G.S.N. Competitive Swarm Optimization Based Unequal Clustering and Routing Algorithms (CSO-UCRA) for Wireless Sensor Networks. *Multimed. Tools Appl.* **2021**, *80*, 26093–26119. [CrossRef]
- 15. Subramani, N.; Mohan, P.; Alotaibi, Y.; Alghamdi, S.; Khalaf, O.I. An Efficient Metaheuristic-Based Clustering with Routing Protocol for Underwater Wireless Sensor Networks. *Sensors* **2022**, *22*, 415. [CrossRef]
- Zhang, Y.; Zhang, Z.; Chen, L.; Wang, X. Reinforcement Learning-Based Opportunistic Routing Protocol for Underwater Acoustic Sensor Networks. *IEEE Trans. Veh. Technol.* 2021, 70, 2756–2770. [CrossRef]
- 17. Yang, J. An Ellipse-Guided Routing Algorithm in Wireless Sensor Networks. Digit. Commun. Netw. 2021. [CrossRef]
- 18. Wang, T.; Qiu, L.; Sangaiah, A.K.; Xu, G.; Liu, A. Energy-Efficient and Trustworthy Data Collection Protocol Based on Mobile Fog Computing in Internet of Things. *IEEE Trans. Ind. Inform.* **2020**, *16*, 3531–3539. [CrossRef]
- 19. Cheng, C.-F.; Li, L.-H. Data Gathering Problem with the Data Importance Consideration in Underwater Wireless Sensor Networks. *J. Netw. Comput. Appl.* **2017**, *78*, 300–312. [CrossRef]
- 20. Ren, J.; Zhang, D.; He, S.; Zhang, Y.; Li, T. A Survey on End-Edge-Cloud Orchestrated Network Computing Paradigms: Transparent Computing, Mobile Edge Computing, Fog Computing, and Cloudlet. ACM Comput. Surv. 2020, 52, 125. [CrossRef]
- 21. Grover, J.; Garimella, R.M. Reliable and Fault-Tolerant IoT-Edge Architecture. In Proceedings of the 2018 IEEE Sensors, New Delhi, India, 28–31 October 2018; pp. 1–4. [CrossRef]
- Wang, T.; Ke, H.; Zheng, X.; Wang, K.; Sangaiah, A.K.; Liu, A. Big Data Cleaning Based on Mobile Edge Computing in Industrial Sensor-Cloud. *IEEE Trans. Ind. Inform.* 2020, 16, 1321–1329. [CrossRef]
- 23. Gong, W.; Qi, L.; Xu, Y. Privacy-Aware Multidimensional Mobile Service Quality Prediction and Recommendation in Distributed Fog Environment. *Wirel. Commun. Mob. Comput.* **2018**, 3075849. [CrossRef]
- Cai, S.; Zhu, Y.; Wang, T.; Xu, G.; Liu, A.; Liu, X. Data Collection in Underwater Sensor Networks Based on Mobile Edge Computing. *IEEE Access* 2019, 7, 65357–65367. [CrossRef]
- 25. Wang, T.; Zhao, D.; Cai, S.; Jia, W.; Liu, A. Bidirectional Prediction-Based Underwater Data Collection Protocol for End-Edge-Cloud Orchestrated System. *IEEE Trans. Ind. Inform.* **2020**, *16*, 4791–4799. [CrossRef]
- de Oliveira, E.M.; Cyrino Oliveira, F.L. Forecasting Mid-Long Term Electric Energy Consumption through Bagging ARIMA and Exponential Smoothing Methods. *Energy* 2018, 144, 776–788. [CrossRef]

- 27. Tayeh, G.B.; Makhoul, A.; Laiymani, D.; Demerjian, J. A Distributed Real-Time Data Prediction and Adaptive Sensing Approach for Wireless Sensor Networks. *Pervasive Mob. Comput.* **2018**, *49*, 62–75. [CrossRef]
- Li, X.; Hu, X.; Zhang, R.; Yang, L. Routing Protocol Design for Underwater Optical Wireless Sensor Networks: A Multiagent Reinforcement Learning Approach. *IEEE Internet Things J.* 2020, 7, 9805–9818. [CrossRef]
- Kim, S. A Better-Performing Q-Learning Game-Theoretic Distributed Routing for Underwater Wireless Sensor Networks. Int. J. Distrib. Sens. Netw. 2018, 14, 1550147718754728. [CrossRef]
- 30. Liu, X.; Yu, J.; Feng, Z.; Gao, Y. Multi-Agent Reinforcement Learning for Resource Allocation in IoT Networks with Edge Computing. *China Commun.* 2020, 17, 220–236. [CrossRef]
- Xiao, L.; Jiang, D.; Chen, Y.; Su, W.; Tang, Y. Reinforcement-Learning-Based Relay Mobility and Power Allocation for Underwater Sensor Networks Against Jamming. *IEEE J. Ocean. Eng.* 2020, 45, 1148–1156. [CrossRef]
- Munir, M.S.; Abedin, S.F.; Hong, C.S. Artificial Intelligence-Based Service Aggregation for Mobile-Agent in Edge Computing. In Proceedings of the 2019 20th Asia-Pacific Network Operations and Management Symposium (APNOMS), Matsue, Japan, 18–20 September 2019; p. 6. [CrossRef]
- Xie, P.; Cui, J.-H.; Lao, L. VBF: Vector-Based Forwarding Protocol for Underwater Sensor Networks. In *Networking 2006: Networking Technologies, Services, and Protocols;* Performance of Computer and Communication Networks; Mobile and Wireless Communications Systems; Boavida, F., Plagemann, T., Stiller, B., Westphal, C., Monteiro, E., Eds.; Springer: Berlin, Germany, 2006; Volume 3976, pp. 1216–1221. ISBN 3-540-34192-7.
- Ismail, A.S.; Wang, X.; Hawbani, A.; Alsamhi, S.; Aziz, S.A. Routing Protocols Classification for Underwater Wireless Sensor Networks Based on Localization and Mobility. *Wirel. Netw.* 2022, 28, 797–826. [CrossRef]
- Boukerche, A.; Sun, P. Design of Algorithms and Protocols for Underwater Acoustic Wireless Sensor Networks. ACM Comput. Surv. 2021, 53, 134. [CrossRef]
- El-Banna, A.A.A.; Wu, K. Machine Learning Modeling for IoUT Networks: Internet of Underwater Things; Springer Nature: Berlin/Heidelberg, Germany, 2021; ISBN 978-3-030-68567-5.
- Ding, Q.; Zhu, R.; Liu, H.; Ma, M. An Overview of Machine Learning-Based Energy-Efficient Routing Algorithms in Wireless Sensor Networks. *Electronics* 2021, 10, 1539. [CrossRef]
- Khan, W.; Wang, H.; Anwar, M.S.; Ayaz, M.; Ahmad, S.; Ullah, I. A Multi-Layer Cluster Based Energy Efficient Routing Scheme for UWSNs. *IEEE Access* 2019, 7, 77398–77410. [CrossRef]
- Chen, Y.; Tang, Y.; Fang, X.; Wan, L.; Tao, Y.; Xu, X. PB-ACR: Node Payload Balanced Ant Colony Optimal Cooperative Routing for Multi-Hop Underwater Acoustic Sensor Networks. *IEEE Access* 2021, 9, 57165–57178. [CrossRef]
- Singhal, S.; Tanwar, S.; Aishwarya; Sinha, A. Ant Colony Optimization Based Routing for Underwater Sensor Network. In Proceedings of the 2020 9th International Conference System Modeling and Advancement in Research Trends (SMART), Moradabad, India, 4–5 December 2020; pp. 33–38. [CrossRef]
- Zhou, Z.; Xing, R.; Duan, Y.; Zhu, Y.; Xiang, J. Event Coverage Detection and Event Source Determination in Underwater Wireless Sensor Networks. *Sensors* 2015, 15, 31620–31643. [CrossRef] [PubMed]
- 42. Zhou, Z.; Fang, W.; Niu, J.; Shu, L.; Mukherjee, M. Energy-Efficient Event Determination in Underwater WSNs Leveraging Practical Data Prediction. *IEEE Trans. Ind. Inform.* 2017, *13*, 1238–1248. [CrossRef]
- 43. Ilyas, N.; Akbar, M.; Ullah, R.; Khalid, M.; Arif, A.; Hafeez, A.; Qasim, U.; Khan, Z.A.; Javaid, N. SEDG: Scalable and Efficient Data Gathering Routing Protocol for Underwater WSNs. In Proceedings of the 6th International Conference on Ambient Systems, Networks and Technologies (Ant-2015), the 5th International Conference on Sustainable Energy Information Technology (Seit-2015); Shakshuki, E., Ed.; Elsevier Science Bv: Amsterdam, The Netherlands, 2015; Volume 52, pp. 584–591.
- 44. Ilyas, N.; Alghamdi, T.A.; Farooq, M.N.; Mehboob, B.; Sadiq, A.H.; Qasim, U.; Khan, Z.A.; Javaid, N. AEDG: AUV-Aided Efficient Data Gathering Routing Protocol for Underwater Wireless Sensor Networks. In Proceedings of the 6th International Conference on Ambient Systems, Networks and Technologies (ANT-2015), the 5th International Conference on Sustainable Energy Information Technology (Seit-2015); Shakshuki, E., Ed.; Elsevier Science Bv: Amsterdam, The Netherlands, 2015; Volume 52, pp. 568–575.
- Akyildiz, I.F.; Pompili, D.; Melodia, T. State-of-the-Art in Protocol Research for Underwater Acoustic Sensor Networks. In Proceedings of the WUWNet 2006—The First ACM International Workshop on Underwater Networks, Los Angeles, CA, USA, 25 September 2006; Volume 2006, pp. 7–16.
- 46. Pottie, G.J.; Kaiser, W.J. Wireless Integrated Network Sensors. Commun. ACM 2000, 43, 51–58. [CrossRef]
- Kao, C.-Y.; Kuo, H.-C.; Chen, J.-W.; Lin, C.-L.; Chen, P.-H.; Lin, Y.-L. RNNAccel: A Fusion Recurrent Neural Network Accelerator for Edge Intelligence. arXiv preprint 2020, arXiv:2010.13311.
- Luong, N.C.; Hoang, D.T.; Gong, S.; Niyato, D.; Wang, P.; Liang, Y.-C.; Kim, D.I. Applications of Deep Reinforcement Learning in Communications and Networking: A Survey. *IEEE Commun. Surv. Tutor.* 2019, 21, 3133–3174. [CrossRef]
- 49. Liu, Z.; Zhu, Z.; Gao, J.; Xu, C. Forecast Methods for Time Series Data: A Survey. IEEE Access 2021, 9, 91896–91912. [CrossRef]
- Hashemi, S.M.; Sanaye-Pasand, M. A New Predictive Approach to Wide-Area Out-of-Step Protection. *IEEE Trans. Ind. Inform.* 2019, 15, 1890–1898. [CrossRef]
- 51. Liu, L.; Wang, Y.; Wang, C.; Ding, F.; Hayat, T. Maximum Likelihood Recursive Least Squares Estimation for Multivariate Equation-Error ARMA Systems. *J. Frankl. Inst.-Eng. Appl. Math.* **2018**, *355*, 7609–7625. [CrossRef]
- Ganjewar, P.D.; Barani, S.; Wagh, S.J. HFBLMS: Hierarchical Fractional Bidirectional Least-Mean-Square Prediction Method for Data Reduction in Wireless Sensor Network. *Int. J. Model. Simul. Sci. Comput.* 2018, 9, 1850020. [CrossRef]

- 53. Hyndman, R.J.; Khandakar, Y. Automatic Time Series Forecasting: The Forecast Package for R. J. Stat. Softw. 2008, 27, 1–22. [CrossRef]
- Mahalakshmi, G.; Sridevi, S.; Rajaram, S. A Survey on Forecasting of Time Series Data. In Proceedings of the 2016 International Conference on Computing Technologies and Intelligent Data Engineering (ICCTIDE'16), Kovilpatti, India, 7–9 January 2016; IEEE: New York, NY, USA, 2016.
- Su, Y.; Gao, W.; Guan, D.; Su, W. Dynamic Assessment and Forecast of Urban Water Ecological Footprint Based on Exponential Smoothing Analysis. J. Clean Prod. 2018, 195, 354–364. [CrossRef]
- 56. Lim, B.; Zohren, S. Time-Series Forecasting with Deep Learning: A Survey. *Philos. Trans. R. Soc. A-Math. Phys. Eng. Sci.* 2021, 379, 20200209. [CrossRef]
- 57. Torres, J.F.; Hadjout, D.; Sebaa, A.; Martinez-Alvarez, F.; Troncoso, A. Deep Learning for Time Series Forecasting: A Survey. *Big Data* **2021**, *9*, 3–21. [CrossRef]